Author's response to reviews

Title: Metabolic alteration of urinary steroids in pre- and post-menopausal women, and men with papillary thyroid carcinoma

Authors:

Man Ho Choi (mh_choi@kist.re.kr)
Ju-Yeon Moon (jy-servant@hanmail.net)
Sung-Hee Choi (sunny@kist.re.kr)
Bong Chul Chung (bcc0319@kist.re.kr)
Eun Jig Lee (ejlee423@yuhs.ac)

Version: 5 Date: 31 July 2011

Author's response to reviews: see over
Dear Editor,

Thank you for your e-mail dated July 30, 2011 on the formatting changes required on our manuscript (ref #: 1840262401525881) entitled; “Metabolic alteration of urinary steroids in pre- and post-menopausal women, and men with papillary thyroid carcinoma”. The manuscript accepted for publication have been formatted based on the changes required.

Thank you in advance for all of your efforts.

Sincerely yours,

Bong Chul Chung, Ph.D.
Principle Investigator
Future Convergence Research Division
Korea Institute of Science and Technology
Tel: (82-2) 958-5067
Fax: (82-2) 958-5059
E-mail: bcc0319@kist.re.kr